Author’s response to reviews

Title: Streptococcus pneumoniae and Haemophilus influenzae in paediatric meningitis patients at Goroka General Hospital, Papua New Guinea: serotype distribution and antimicrobial susceptibility in the pre-vaccine era

Authors:

Andrew Greenhill (andrew.greenhill@federation.edu.au)
Suparat Phuanukoonnnon (phuanukoonnnon@wehi.edu.au)
Audrey Michael (audrey.michael@pngimr.org.pg)
Mition Yoannes (mition.yoannes@pngimr.org.pg)
Tilda Orami (tilda.orami@pngimr.org.pg)
Helen Smith (Helen.Smith4@health.qld.gov.au)
Denise Murphy (denisemarymurphy@gmail.com)
Christopher Blyth (Christopher.blyth@uwa.edu.au)
John Reeder (reederj@who.int)
Peter Siba (peter.siba@pngimr.org.pg)
William Pomat (william.pomat@pngimr.org.pg)
Deborah Lehmann (Deborah.Lehmann@telethonkids.org.au)

Version: 2 Date: 03 Oct 2015

Author’s response to reviews:

Minor changes have been made as requested by the editor.